Author's response to reviews

Title: How Personal and Standardized Coordination Impact Implementation of Integrated Care

Authors:

Justin K Benzer (justin.benzer@va.gov)
Irene E Cramer (irene.cramer21@gmail.com)
James F Burgess Jr (james.burgess@va.gov)
David C Mohr (david.mohr2@va.gov)
Jennifer L Sullivan (Jennifer.sullivan@va.gov)
Martin P Charns (martin.charns@va.gov)

Version: 6 Date: 6 August 2015

Author's response to reviews:

Added more detail to the Author Contributions section so the role of each author is listed.

Moved the heading 'Background' to the start of the paper (pg 3, line 51), added conceptual framework header in place of 'Background", and added keywords sub section after the Abstract.